

EO-1 Weekly Report


	Earth Observing – 1

	June 24, 2015 – June 30, 2015

	DOY
	175
	–
	181

	Mission Day
	5379
	–
	5385


EO-1 SPACECRAFT

Instruments

· Scheduled 129 science Data Collection Events (DCEs)

· Received all DCEs

· Performed Instrument Calibration 

· Conducted ALI Internal Calibration 1 at 180/00:00:00
EO-1 MISSION OPERATIONS CENTER

Real-Time

· Maintaining nominal spacecraft state of health 

· Issued a command to jam the Leap Second to 17 seconds. 

· Uploaded 7 degree and 28 degree Nominal and Slow HSI only Lunar Calibrations

· The goal file that the dump failed during the pass on USAK03 at 176/1956z pass was not a problem with the dump or goal file itself. That particular goal file was for DOY 178, and the problems occurred on 176 at end of day. ASE does not know about a goal file or try and process it until it is committed, which basically means we give ASE a pointer to that file location and tell it that it exists. Since we were operating on the DOY 176 and DOY 177 goal files at the time of the problem, ASE had no idea there was even an issue so no scenes or image dumps were affected.  Also, that goal file for DOY 178 was uploaded on the USAK03 pass at 177/0713z and it was re-dumped, verified, committed, and accepted fine so the problem seems like a one time event and not a problem with the goal file itself. The command to re-dump the goal file was sent to EO-1 and accepted. Then when it was telling the SM which data location is holding the newly uploaded goal file it was told to dump back down, there was an overrun in the pipe and some of that command seems to have been lost between the point where EO-1 accepted the command and the SM was executing the command. The station where the dump failed at was on USAK03 at 176/1956z, but I do not believe it was a problem with the station at all since the command was sent to EO-1 and accepted. It was the inner workings of EO-1 that failed in this instance. Also the retry of that goal file was also on USAK03 at 0713z and worked nominally.

· We got a TDRS pass, issued the WSMABORTD command, with all the dumps commanded to abort, the system was nominal. 

· Changed scripts from WGS to WG1 according to ground station name change

· Looking into supporting more lunar calibrations

· Taking USAK03 Supports

· Successfully Taking S-band supports

· Successfully Taking X-band supports

· Successfully using tracking data from USAK03

· Gathering data from GPS to use as tracking data input for STK

· Changed the EOP script on real-time machines to take into account the future leap second broadcast from the GPS satellites.
· Leap Second change in July.  

· Observing Star Tracker data to view cold/hot pixels 

Mission Planning
· Planned and Created 7 degree and 28 degree Nominal and Slow HSI only Lunar Calibrations

· Changed scripts from WGS to WG1 according to ground station name change

· Rewriting the Lunar Calibration procedure and testing the changes

· The procedure has been written, testing to ensure the correctness of the procedure

· Rewriting the TDRS contact procedure and testing the changes

· First set of tests worked well

· Working with ASPEN developers to configure various Lunar Calibration sequences 

· Bringing back Modified Lunar Calibrations
Flight Dynamics 
· Planned and Created 7 degree and 28 degree Nominal and Slow HSI only Lunar Calibrations

· Changed scripts from WGS to WG1 according to ground station name change

· USAK03 tracking data is working nominally
· Testing using GPS based tracking data for Orbit Determination

· Creating an Everyday FD orbit determination scenario on the backup FD machine.  

· Verifying rough draft of delta-v planning procedures

· Rewriting the Lunar Calibration procedure and testing the changes

· Rewriting the TDRS contact procedure and testing the changes

· First set of tests worked well

· Researching Modified Lunar Calibrations to be resumed once MP is ready for them
Central File Hub
· Adjusted scripts to deliver data to new JPL machine

· One week of successful testing has completed

· Phase one tests are ready for outside entity delivery.

· SGS and WGS tracking data are waiting for a break in WALLOPS activities to begin testing.

· Phase 2 FD scripts have no changes from Phase 1 scripts
· All non-FD scripts are implemented Phase 2 

· Beginning Phase 3 work on deliveries for the central file hub

· Central file hub delivery structure set
· Passwordless ssh set up from FD to eo1out and eo1in to and from eo1out
· Script to put FD products onto file hubs is complete and ready for use
· Script to sort FD products into destination directories is complete and ready for use

· Base delivery script completed

· Designing delivery code architecture

· Reviewing current deliveries for products, emails, and destinations

· Delivery will be completed via FD computers as is the current standard
System Administration

· All MOC linux systems except dps04, fluflu2, gateway and gateway2 have had the bash patch applied.

· Awaiting test of patched dps05 and judo2

· Need a schedule for credentialed scan

· Attended RHEL baseline review meeting

· Attended security tag-up and FPD bi-weekly meeting

· Completed test build of gateway2.  Some testing still needed

· Reinstalled SEP in eo1aspen32

· Relocated external drive on dps04

· Created backups for the ASIST computers, flight dynamics computers, ASPEN computers, countdown clock, data processing computers, FEDS computers, firewall, trending system, regional logs server, and a full backup of the MOC CNE computer.

GROUND AND SPACE NETWORK

Station Downtimes 

· HGS is down with no estimated uptime
Operational Discrepancies:

· None to report at this time


[image: image1.emf]Date DOY S/X-Band S-Band Only Non-DSMC TDRSS Daily Totals

24-Jun-15 175 7 2 0 0 9

25-Jun-15 176 5 4 1 0 10

26-Jun-15 177 5 2 0 0 7

27-Jun-15 178 7 4 1 0 12

28-Jun-15 179 6 3 1 0 10

29-Jun-15 180 5 4 1 0 10

30-Jun-15 181 2 6 1 0 9

37 25 5 0 67 Weekly Totals


UPCOMING EVENTS
· Planning and performing off-angle lunar calibrations

· Continuation of Phase 1/2 FD, waiting on WALLOPS support 
· Moving on to Phase 3 for central file hub, deliveries
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		Date		DOY		S/X-Band		S-Band Only		Non-DSMC		TDRSS		Daily Totals

		24-Jun-15		175		7		2		0		0		9

		25-Jun-15		176		5		4		1		0		10

		26-Jun-15		177		5		2		0		0		7

		27-Jun-15		178		7		4		1		0		12

		28-Jun-15		179		6		3		1		0		10

		29-Jun-15		180		5		4		1		0		10

		30-Jun-15		181		2		6		1		0		9

		Weekly Totals				37		25		5		0		67
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